Author’s response to reviews

Title: A framework for the ethical assessment of chimeric animal research involving human neural tissue

Authors:

Sebastian Porsdam Mann (sebastian_mann@hms.harvard.edu)
Rosa Sun (rosa_sun@hotmail.co.uk)
Goran Hermeren (Goran.hermeren@med.lund.se)

Version: 4 Date: 28 Nov 2018

Author’s response to reviews:

Thank you for your time. We have addressed all questions in the response to the editor document. Regarding authorship, we would like to make clear that all three authors have contributed equally throughout the entire project.